Author's response to reviews

Title: Vitamin B6 deficiency and diseases in elderly - a study in nursing homes

Authors:

Ida K Kjeldby (ida.k.kjeldby@gmail.com)
Gunvor S Fosnes (gunvor.fosnes@gmail.com)
Solveig C Ligaarden (solveigligaarden@sykehuset-innlandet.no)
Per G Farup (per.farup@ntnu.no)

Version: 5 Date: 24 November 2012

Author's response to reviews: see over
The Editor,
BMC Geriatrics

Dear Editor,

Thanks again for your evaluation of our manuscript “Vitamin B6 deficiency and diseases in elderly – a study in nursing homes”. Hereby follows an itemized, point-by-point response to the comments.

**Comments to reviewer 2: Jessica Sautter**

Minor essential revisions

1. The sentence proposed by the reviewer has been added to the section “Strengths and limitations”.
2. “Predictor” has been replaced by “were associated with” in the Discussion.

Some language corrections have been made.

We kindly ask you to evaluate the revised version of the paper for publication in BMC Geriatrics.

We thank you in advance.

Sincerely yours

On behalf of the authors

Per G Farup MD, Ph.D
Professor, Faculty of Medicine, Norwegian University of Science and Technology.